Author's response to reviews

Title: The knowledge and attitudes of breast self-examination and mammography in a group of women in a rural area in western Turkey

Authors:

Pinar Erbay Dundar (pinarerbaydundar@yahoo.com)
Dilek Ozmen (ozmendilek@yahoo.com)
Beyhan Ozyurt (beyhanozyurt@hotmail.com)
Gokce Haspolat (gokcemmdr@hotmail.com)
Filiz Akyildiz (dr.filizakyildiz@hotmail.com)
Sumeyra Coban (dr.sumeyra@hotmail.com)
Gamze Cakiroglu (drgamzecak@hotmail.com)

Version: 6 Date: 23 February 2006

Author's response to reviews:

This is the version of copyediting of the manuscript.